Author's response to reviews

**Title:** The effect of cartilage and bone density of mushroom-shaped, photooxidized, osteochondral transplants: an experimental study on graft performance in sheep using transplants originating from different species

**Authors:**
- Anja C Waselau (anja.waselau@gmx.net)
- Daniel Nadler (d_nadler@gmx.ch)
- Jessika-M. V. Muller (jmA¼ller@vetclinics.unizh.ch)
- Katalin Zlinszky (zlin@vetpath.unizh.ch)
- Monika Hilbe (hilbe@vetpath.unizh.ch)
- Jorg A Auer (jauer@vetclinics.unizh.ch)
- Brigitte von Rechenberg (bvonrechenberg@vetclinics.unizh.ch)

**Version:** 4  **Date:** 25 November 2005

**Author's response to reviews:**

Tab. in the text have been changed to see "additional files"
Figures 3a and 3b have been merged to Fig.3
Figures 5a and b have been merged to Fig. 5